Author’s response to reviews

Title: Time to viral load suppression and its associated factors in cohort of patients taking antiretroviral treatment in East Shewa Zone, Oromiya, Ethiopia, 2018

Authors:

Jemal Ali (zaweyb@yahoo.com)

Version: 2 Date: 01 Nov 2019

Author’s response to reviews:

I have tried to address comment by the reviewer. I have tried to go throught all the document and make a total of 47 ammendments/corrections. These includes 3 capitalization, 15 Grammar, 25 addition/deletion of words such as "the, an, a" and 4 sentences re written.

for specific reviewer comment

1. Baseline CD4 count is the value of CD4 count of the study participants". What do you mean? I have corrected it as"it is the CD4 count of the participants before starting ART"

2. Baseline PVL count in which the patients plasma viral load was measured at the start of the study, was taken when the study participant enrolled in the study" corrected as 'Baseline PVL count in this study is defined as the plasma viral load measured when participant enrolled in the stud.